Author's response to reviews

Title: Assessment of need for Inpatient Treatment for Mental Disorder among Female Prisoners: A Cross-Sectional Study of Provincially Detained Women in Ontario

Authors:

Roland Jones (roland.jones@camh.ca)
Kiran Patel (kiran.patel@camh.ca)
Alexander Simpson (sandy.simpson@camh.ca)

Version: 4 Date: 12 Mar 2019

Author's response to reviews:

Many thanks for requesting a further revision.

We have proof read the document as requested and made some minor corrections and addressed some errors in the punctuation and spacing.